Author's response to reviews

Title: Serum procalcitonin and CRP levels in non-alcoholic fatty liver disease: a case control study

Authors:

Nevin Oruc (nevintr@yahoo.com)
Omer Ozutemiz (omer.ozutemiz@ege.edu.tr)
Gul Yuce (gulyuce@ege.edu.tr)
Ulus S Akarca (ulus.akarca@ege.edu.tr)
Galip Ersoz (galip.ersoz@ege.edu.tr)
Fulya Gunsar (fulya.gunsar@ege.edu.tr)
Yucel Batur (yucelbatur@yahoo.com)

Version: 6 Date: 5 February 2009

Author's response to reviews:

-HOMA-R is corrected as HOMA-IR under the last table.